Author's response to reviews

Title: MRI findings are often missed in the diagnosis of Creutzfeldt-Jakob disease

Authors:

Christopher J Carswell (mrcarswell@hotmail.com)
Andrew G Thompson (a.thompson@prions.ucl.ac.uk)
Anna A Lukic (a.lukic@prions.ucl.ac.uk)
John M Stevens (john.stevens@uclh.nhs.uk)
Peter Rudge (p.rudge@prions.ucl.ac.uk)
Simon H Mead (s.mead@prions.ucl.ac.uk)
John Collinge (pacollinge@prions.ucl.ac.uk)
Harpreet Hyare (h.hyare@prions.ucl.ac.uk)

Version: 2 Date: 14 August 2012

Author's response to reviews: see over
14\textsuperscript{nd} August 2012

Dear Editor of BMC Neurology,

We have revised the paper as per the request of the editorial team prior to formal review. Specifically we first include a statement in the methods section that confirms that the patients gave full consent to review by the National Prion Clinic and were enrolled in clinical therapeutic or observational trials which were carried out in accordance with the Helsinki declaration.

Secondly we were asked to further anonymize the supplementary figure 2 legend. Now all case ages and historical details have been removed and replaced only by their WHO diagnostic category. For example: A) a “probable” sCJD case.

We hope that these changes are sufficient and look forward to your thoughts and opinions regarding the paper.

Sincerely,

Chris Carswell & Harpreet Hyare